Author’s response to reviews

Title: Modified Entropy-Based Procedure Detects Gene-Gene-Interactions in Unconventional Genetic Models

Authors:

Jörg Malten (joerg.malten@student.uni-luebeck.de)
Inke König (inke.koenig@uni-luebeck.de)

Version: 2 Date: 11 Mar 2020

Author’s response to reviews:

1. Please include References and citations in the next revision: they are currently missing.

We apologize; now all references and citations are included.

2. Please change the article type to “Technical Advance” and ensure it is formatted in the following manner:
   https://bmcmedgenomics.biomedcentral.com/submission-guidelines/preparing-your-manuscript/technical-advance-article

We changed the article type accordingly.

3. Please transfer the sentence “We would like to thank…” in the Funding section, to the Acknowledgements.

We moved the sentence accordingly.

4. Please provide more details regarding the specific role of each of the funding bodies in your study. If the funding bodies only provided the financial means to allow the authors to carry out the study, please state “The funding bodies played no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript” and include this statement in the Funding section of your manuscript.

We added the information that the funding bodies played no role.